\documentclass[11pt]{article}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{epstopdf}
\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `dirname #1`/`basename #1 .tif`.png}

\textwidth = 6.5 in
\textheight = 9 in
\oddsidemargin = 0.0 in
\evensidemargin = 0.0 in
\topmargin = 0.0 in
\headheight = 0.0 in
\headsep = 0.0 in
\parskip = 0.2in
\parindent = 0.0in

\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{definition}{Definition}

\begin{document}

\begincenter{}\endcenter







\centerline{\bf MATHEMATICS 366, SPRING 1998 }
\centerline{Lecturer: Qing HAN} 
\centerline{Office: Math Building 215}
\centerline{Phone: 631-6834}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\vskip20pt


\medskip
\title{TEXTBOOK}
\smallskip
\begingroup\obeylines
\temp{\
& Principles of Mathematical Analysis, 3rd Edition, by W. Rudin    
\cr }\endgroup

\medskip
\title{CHAPTERS TO BE COVERED}
\smallskip
\begingroup\obeylines
\temp{\ & This is the continuation of Math365 in the fall semester. In this semester we will finish the rest of the textbook. We will not follow the textbook section by section, although all homework will be chosen from the textbook.  
\cr 
\hfill  Chapter 2  
& Basic Topology    
\cr 
\hfill  Chapter 7 
& Sequences and Series of Functions   
\cr
\hfill  Chapter 8
& Some Special Functions    
\cr
\hfill  Chapter 9
& Functions of Several Variables   
\cr
\hfill  Chapter 10   
& Integration of Differential Forms      
\cr     }\endgroup

\medskip
\title{HOMEWORK}
\smallskip
\begingroup
\temp{\
& The homework of each week will be due the following {\bf Tuesday}.  
\cr     
& Homeworks more than 7 days late will not be accepted. 
\cr}\endgroup


\medskip
\title{TESTS AND FINAL EXAM}
\smallskip
\begingroup\obeylines
\temp{\
& There will be two 1-hour tests and a final exam.  
\cr 
& The schedule for two tests will be announced in class. 
\cr}\endgroup

\medskip
\title{GRADING POLICY }
\smallskip
\begingroup
\temp{\
& Final grades will be determined as follows: 25\% of each test, 10\% of homework and 40\% of final exam. 
\cr     }\endgroup

\vfill\eject

\medskip

 \end{document}

 
